Author's response to reviews

Title: Quality assessment of systematic reviews or meta-analyses of nursing interventions conducted by Korean reviewers

Authors:

Hyun-Ju Seo (shj5th@korea.ac.kr)
Kyeong Uoon Kim (kukim@kangwon.ac.kr)

Version: 7 Date: 20 July 2012

Author's response to reviews: see over
Dear, Dr. Adrian Aldcroft

Our manuscript was edited by a native-English speaker to improve readability. And we carefully corrected the manuscript to conform to the journal style. In addition, we revised the title to “Quality assessment of systematic reviews or meta-analyses of nursing interventions conducted by Korean reviewers”.

We attached an editing certificate as follows:

We hope that the revised version of our paper is now suitable for publication in the BMC medical research methodology.

Yours sincerely,
Kyeong Uoon Kim (RN, MPH, PhD)